Author's response to reviews

Title: Associated factors for treatment delay in pulmonary tuberculosis in HIV-infected individuals: a nested case-control study

Authors:

Isabella Coimbra (isabella.coimbra@uol.com.br)
Magda Maruza (magdamaruza@yahoo.com.br)
Maria de Fatima PM Albuquerque (militaofatima@gmail.com)
Libia V Moura (mouralibia@oi.com.br)
George TN Diniz (george@cpqam.fiocruz.br)
Democrito B Miranda Filho (demofilho@uol.com.br)
Heloisa R Lacerda (helramos@terra.com.br)
Laura C Rodrigues (laura.rodrigues@lshtm.ac.uk)
Ricardo AA Ximenes (ricardo.ximenes@pq.cnpq.br)

Version: 6 Date: 28 May 2012

Author's response to reviews:

Follows, attached cover letter and responses/comments to the referees with all the changes made in the article.